Author's response to reviews

Title: Cardiac condition during cooling and rewarming periods of therapeutic hypothermia after cardiopulmonary resuscitation

Authors:

Serdar Demirgan (serdardemirgan@hotmail.com)
Kerem Erkalp (keremerkalp@hotmail.com)
M. Salih Sevdi (salihseydi@yahoo.com)
Meltem Turkay Aydogmus (meltem72_3@hotmail.com)
Numan Kutbay (nkutbay@hotmail.com)
Aydin Firincioglu (aydinfirincioglu@gmail.com)
Ali Ozalp (ali_ozalp3@hotmail.com)
Aysin Alagol (aysinalagol@yahoo.com)

Version: 7  Date: 5 August 2014

Author's response to reviews:

Dear Editor;

Email addresses for all authors were included on title page.

We removed the tables as additional files and incorporated them into the manuscript, after the references.

We removed our cover letter and title page as additional files in the system. Cover letters had been uploaded on the submission system and the title page has been the first page of the manuscript.

Sincerely yours.